Author’s response to reviews

Title: Common variable immunodeficiency syndrome with right aortic arch; a case report

Authors:
Dr Didem A Yalcin (adidyal@yahoo.com)
Nese Aydemir (naydemir@pamukkale.edu.tr)
Huseyin Turgut (hturgut@pamukkale.edu.tr)
Dr Hakan R Erbay (rherbay@pamukkale.edu.tr)
Nevzat A Yalcin (anyalcin@akdeniz.edu.tr)

Version: 4 Date: 23 Jan 2004

Emma Parkin 23rd, January 2004
Editorial Administrator,
BMC Journals

We revised our manuscript according to your and referees’ suggestions as follows:

Manuscript sections formatted as follows: Abstract; Background; Case Presentation; Discussion; Conclusions; Acknowledgements; References; Figure legends.
Background section is added.

Reviewer (Rafal Pawliczak):
The manuscript was reviewed for correction of linguistic and grammar mistakes. We corrected some editing errors in the text.

Sincerely

A. Didem Yalcin, MD.